Author’s response to reviews

Title: Implementing Collaborative Care to reduce depression for rural Native American/Alaska Native people

Authors:
Deborah Bowen (dbowen@uw.edu)
Diane Powers (powersd@uw.edu)
Joan Russo (jerusso@uw.edy)
Rob Arao (araor@uw.edu)
Erin Lepoire (lepoire2@uw.edu)
Earl Sutherland (earl.sutherland@bighornvalley.org)
Anna Ratzliff (annar22@uw.edu)

Version: 4 Date: 17 Dec 2019

Author’s response to reviews:
I changed the title text last round to match your critique. It contains Native American instead of American Indian.